Author’s response to reviews

Title: Genotoxicity Evaluation of So-ochim-tang-gamibang (SOCG), a Herbal Medicine

Authors:

Mi Young Lee (mylee@biotoxtech.com)
Yang-Chun Park (omdpyc@dju.ac.kr)
Mirim Jin (mirimj@gachon.ac.kr)
Eunseok Kim (mybestcando@naver.com)
Jeong June Choi (c27022@dju.ac.kr)
In Chul Jung (npjeong@dju.ac.kr)

Version: 2 Date: 27 Dec 2017

Author’s response to reviews:

Przemysław Łukasz Mikołajczak (Reviewer 1): I think that the answers are acceptable and I congratulate Authors a good paper. However, there is only 1 small mistake in the references, since still the Ref. 19 (previous 13) is not correct and should be replaced from:


Into:


Answer> We appreciate reviewer’s point. We corrected the mistake according to reviewer’s suggestion. (References section, line 18, page 20, Ref. 19)

Angel Treasa Alex, Ph.D. (Reviewer 2): There are no existing reports of adverse effects or toxicity of SOCG. Since the Genotoxicity Evaluation of So-ochim-tang-gamibang (SOCG) was carried out NOT in response to the reports on similar properties of this drug, the research gap/objective of this work may be justified in the introduction.
Answer: Thank you very much Dr. Angel Treasa Alex. Your last comments helped us to improve our study.